Author's response to reviews

Title: Remote Noninvasive Allograft Rejection Monitoring for Heart Transplant Recipients: Study Protocol for the Novel Evaluation With Home Electrocardiogram And Remote Transmission (NEW HEART) Study

Authors:

Lynn V Doering (ldoering@sonnet.ucla.edu)
Kathleen Hickey (kth6@columbia.edu)
David Pickham (david.pickham@ucsf.edu)
Belinda Chen (bchen@sonnet.ucla.edu)
Barbara J Drew (barbara.drew@ucsf.edu)

Version: 2 Date: 11 January 2012

Author's response to reviews:

spelling error - change "allograph" to "allograft." Correct title should read:

Remote Noninvasive Allograft Rejection Monitoring for Heart Transplant Recipients: Study Protocol for the Novel Evaluation With Home Electrocardiogram And Remote Transmission (NEW HEART) Study

Also corrected error in manuscript.